



BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242

Fax: +1-925-223-8243

E-mail: bpgoffice@wjgnet.com

<http://www.wjgnet.com>

ANSWERING REVIEWERS

Name of journal: World Journal of Orthopedics

ESPS manuscript NO: 29850

Title: Growing spine deformities: are magnetic rods the final answer?

Reviewer's code: 00505427

COMMENTS TO AUTHORS

Authors should be congratulated on a comprehensive historical overview and technology update on growing rods for EOS.

Reviewer's code: 00501328

COMMENTS TO AUTHORS

The work is interesting and well written.

AUTHORS RESPONSE

Thank you very much to both the reviewers. As per their review, no revision is required. We have corrected the referencing format as suggested by the Editor.

Dr. Ashok Johari

Dr. Amit Nemade

24th Nov. 2016